Title: Nystagmus as an early ocular alteration in Machado-Joseph disease (MJD/SCA3)

Authors:

Mafalda Raposo (msraposo@uac.pt)
João Vasconcelos (neurazor@hdes.pt)
Conceição Bettencourt (c.bettencourt@ucl.ac.uk)
Teresa Kay (teresakay8@gmail.com)
Paula Coutinho (coutinho.mpaula@gmail.com)
Manuela Lima (mlima@uac.pt)

Version: 2 Date: 12 November 2013

Author's response to reviews:

Changes performed in initial submission step:

1. The title 'Introduction' was changed to 'Background'.

2. The following section "Conclusions" was added.

3. The Acknowledgements section was in the correct place in the manuscript (before references), in the first version submitted.